Author's response to reviews

Title: Risk-adjusted operative delivery rates and maternal-neonatal outcomes as measures of quality assessment in obstetric care. A multicenter prospective study.

Authors:

Gianpaolo Maso (gianpaolo.maso@burlo.trieste.it)
Lorenzo Monasta (lorenzo.monasta@burlo.trieste.it)
Monica Piccoli (picciuli72@gmail.com)
Luca Ronfani (luca.ronfani@burlo.trieste.it)
Marcella Montico (marcella.montico@burlo.trieste.it)
Francesco De Seta (fradeseta@gmail.com)
Sara Parolin (parolinsara@gmail.com)
Caterina Businelli (catebusinelli@gmail.com)
Laura Travan (laura.travan@burlo.trieste.it)
Salvatore Alberico (salvatore.alberico@burlo.trieste.it)

Version: 5 Date: 14 January 2015

Author's response to reviews:

Dear Editor-in-chief,

As corresponding author of the manuscript numbered 2617435751191380 "Risk-adjusted operative delivery rates and maternal-neonatal outcomes as measures of quality assessment in obstetric care. A multicenter prospective study", on behalf of the co-authors, I submitted a copy of the revised version of the manuscript according to the suggested changes in title page (all authors’ suffixes have been removed ) and authors’ contribution.

As corresponding author I confirm that all authors read and approved the final manuscript and agree with its submission to BMC- Pregnancy and Childbirth.

As corresponding author I also state that all the co-authors Lorenzo Monasta, Monica Piccoli, Luca Ronfani, Marcella Montico, Sara Parolin, Caterina Businelli, Laura Travan and Salvatore Alberico did agree to include Francesco De Seta into the authorship.

In submitting the revised manuscript to any of the journals published by BioMed Central I certify that:

1. I am authorized by my co-authors to enter into these arrangements.
2. I warrant, on behalf of myself and my co-authors, that:
   a. the article is original, has not been formally published in any other peer-reviewed journal, is not under consideration by any other journal and does not infringe any existing copyright or any other third party rights;
   b. I am/we are the sole author(s) of the article and have full authority to enter into
this agreement and in granting rights to BioMed Central are not in breach of any other obligation.

c. the article contains nothing that is unlawful, libellous, or which would, if published, constitute a breach of contract or of confidence or of commitment given to secrecy;

d. I/we have taken due care to ensure the integrity of the article. To my/our - and currently accepted scientific - knowledge all statements contained in it purporting to be facts are true and any formula or instruction contained in the article will not, if followed accurately, cause any injury, illness or damage to the user.

e. I agree to BioMed Central's Open Data policy

3. I, and all authors, agree that the article, if editorially accepted for publication, shall be licensed under the Creative Commons Attribution License 4.0. If the law requires that the article be published in the public domain, I/we will notify BioMed Central at the time of submission upon which the article shall be released under the Creative Commons 1.0 Public Domain Dedication waiver.

Please let me know of your decision at your earliest convenience.

With my best regards,

Sincerely yours

Gianpaolo Maso MD

Department of Obstetrics and Gynecology

Institute for Maternal and Child Health – IRCCS Burlo Garofolo

Via dell’Istria 65/1, Trieste 34137, Italy

Phone at work: +39 040 3785303/307 - Personal Mobile: +39 335 6114569

E-mail address: gianpaolo.maso@burlo.trieste.it